Author's response to reviews

Title: Use of a T cell interferon gamma release assay in the investigation for suspected active tuberculosis in a low prevalence area

Authors:

Niclas Winqvist (Niclas.Winqvist@skane.se)
Per Björkman (Per.Bjorkman@med.lu.se)
Ann Norén (Ann.Noren@skane.se)
Håkan Miörner (Hakan.Miorner@skane.se)

Version: 4 Date: 18 June 2009

Author's response to reviews:

The requested formatting changes have been made i.e.
- the table vertical elements have been removed
- the tables have been renumbered in ascending numerical order 1-6
- references to the tables in the text have been updated to reflect the changes